\begin{table}%T1
\caption{Relative (dimensionless) O~III transition probability values. $A^{\rm rel}_{\exp}$
represent our experimental values  (with estimated accuracies) related to the
reference  $A$ value of the 326.085~nm O~III transition.
 $A_{\rm N}^{\rm rel}$  denote  transition probability values tabulated by NIST (\cite{NIST03}). $A_{\rm B}^{\rm rel}$
 (Berg et al. \cite{Berg64}; Berg \cite{Berg67}) and $A_{\rm TK}^{\rm rel}$ (Truhan \& Kiseljevskij \cite{Truhan67})
 represent experimentally obtained $A^{\rm rel}$ values using the RLIR method.
 $A^{\rm rel}_{\rm P}$ (Pinnington et al. \cite{Pinnington70}) denote relative $A$ values obtained on the basis of the
 beam-foil technique. $A_{\rm A}^{\rm rel}$ (Aggarwal et al. \cite{Aggarwal97}); $A_{\rm BK}^{\rm rel}$ (Bhatia \& Kastner \cite{Bhatia93})
 and $A_{\rm FF}^{\rm rel}$ (Froese Fischer \cite{Froese94}) represent recently calculated transition
 probability values. Data in brackets denote absolute
 $A$ values of the reference 326.085~nm transition (in $10^8$ s$^{-1}).$ $J_{\rm f}$ and $J_{\rm i}$ are the
 inner quantum numbers of the final (f) and initial (i) state of the transition.
  Atomic data are taken from NIST (\cite{NIST03}).}\label{tab1}

  \vskip 1mm
%%\centerline

{\vbox{\offinterlineskip \hrule\vskip0.4mm\hrule\halign{&\vrule#&
\strut\hbox to1.55pt{}\hfil#\hbox to1.55pt{}\cr
%  \strut\hbox to1pt{}\hfil#\hbox to1pt{}\cr
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&Multiplet\hfil&&$\lambda$\hfil&&$E_i$\hfil&&$A_{\exp}^{\rm rel}$\hfil&&$A_{\rm N}^{\rm rel}$\hfil&&$A_{\rm B}^{\rm rel}$\hfil
&&$A_{\rm TK}^{\rm rel}$\hfil&&$A_{\rm A}^{\rm rel}$\hfil&&$A_{\rm BK}^{\rm rel}$\hfil&&$A_{\rm FF}^{\rm rel}$\hfil&&$A_{\rm P}^{\rm rel}$\hfil&\cr
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$J_{\rm f} - J_{\rm i}$\hfil&&(nm)\hfil&&(eV)&& && && && && && && && &\cr
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
\noalign{\hrule depth.3pt}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3s\ {\rm ^3P^0} - 3p\ {\rm ^3D}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&0 $-$ 1\hfil&&375.723&&36.43&&0.36$~\pm~$13\%&&0.331&&0.31\hfil
&&\hfil&&0.280\hfil&&0.253&&0.324 && \hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 1\hfil&&377.402&&36.43&&0.26$~\pm~$15\%&&0.233&&0.23\hfil
&&\hfil&&0.201\hfil&&0.182&&0.230 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 2\hfil&&375.469&&36.45&&0.45$~\pm~$15\%&&0.448&&0.44\hfil
&&0.61\hfil&&0.379\hfil&&0.343&& 0.438 &&0.607&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 2\hfil&&379.127&&36.45&&0.12$~\pm~$15\%&&0.133&&\hfil
&&\hfil&&0.117\hfil&&0.107&&0.132 &&0.197&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 3\hfil&&375.987&&36.48&&0.51$~\pm~$13\%&&0.583&&0.48\hfil
&&0.67\hfil&&0.497\hfil&&0.452&&0.571 &&0.815&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3s\ {\rm ^3P^0} - 3p\ {\rm ^3S}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&0 $-$ 1\hfil&&329.938&&36.89&&0.11$~\pm~$12\%&&0.098&&0.07\hfil
&&\hfil&&0.086\hfil&&0.069&&0.090 && \hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 1\hfil&&331.232&&36.89&&0.25$~\pm~$12\%&&0.274&&0.25\hfil
&&\hfil&&0.245\hfil&&0.210&& 0.250&&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 1\hfil&&334.076&&36.89&&0.35$~\pm~$12\%&&0.391&&\hfil
&&\hfil&&0.359\hfil&&0.305&&0.345 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3s\ {\rm ^3P^0} - 3p\ {\rm ^3P}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 1\hfil&&303.541&&37.23&&0.28$~\pm~$12\%&&0.273&&\hfil
&&\hfil&&0.316\hfil&&0.353&&0.256 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 1\hfil&&305.927&&37.23&&0.50$~\pm~$11\%&&0.519&&\hfil
&&\hfil&&0.568\hfil&&0.610&&0.487 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 2\hfil&&302.342&&37.25&&0.27$~\pm~$11\%&&0.285&&\hfil
&&\hfil&&0.321\hfil&&0.355&&0.261 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 2\hfil&&304.710&&37.25&&0.87$~\pm~$11\%&&0.887&&\hfil
&&0.64\hfil&&0.986\hfil&&1.08\hfil&&0.823 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3s\ {\rm ^1P^0} - 3p\ {\rm ^1D}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 2\hfil&&298.378&&38.01&&1.15$~\pm~$10\%&&1.28\hfil&&\hfil
&&1.06\hfil&&1.35\hfil&&1.15\hfil&& &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3p\ {\rm ^3D} - 3d\ {\rm ^3F^0}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 2\hfil&&326.720&&40.23&&0.89$~\pm~$6\%\hfil&&0.940&&0.75\hfil
&&\hfil&&0.800\hfil&&0.905&&0.596 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 2\hfil&&328.183&&40.23&&0.18$~\pm~$8\%\hfil&&0.172&&0.10\hfil
&&\hfil&&0.132\hfil&&0.160&&0.111 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&\omit&&\omit&&\omit&&\omit&&(1.68)&&(2.45)\hfil
&&(1.82)&&(1.217)&&(2.19)&&(1.98) &&(1.78)\hfil&\cr
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 3\hfil&&326.085&&40.25&&1.00$~\pm~$3\%\hfil&&1.00\hfil&&1.00\hfil
&&1.00\hfil&&1.00\hfil&&1.00\hfil&&1.00 &&1.00\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&3 $-$ 3\hfil&&328.445&&40.25&&0.17$~\pm~$8\%\hfil&&0.123&&0.13\hfil
&&\hfil&&0.113\hfil&&0.111&&0.103 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&3 $-$ 4\hfil&&326.532&&40.27&&1.11$~\pm~$6\%\hfil&&1.12\hfil&&1.18\hfil
&&0.94\hfil&&1.11\hfil&&1.11\hfil&& &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3p\ {\rm ^1P} - 3d\ {\rm ^1D^0}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 2\hfil&&295.969&&40.26&&0.75$~\pm~$12\%&&1.09\hfil&&\hfil
&&\hfil&&0.884\hfil&&1.04\hfil&&0.682 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3p\ {\rm ^3P} - 3d\ {\rm ^3D^0}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 1\hfil&&371.402&&40.57&&0.24$~\pm~$13\%&&0.242&&0.24\hfil
&&\hfil&&0.216\hfil&&0.155&&0.219 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&1 $-$ 2\hfil&&370.727&&40.58&&0.47$~\pm~$7\%\hfil&&0.437&&0.50\hfil
&&\hfil&&0.419\hfil&&0.310&&0.428 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 3\hfil&&371.508&&40.58&&0.59$~\pm~$8\%\hfil&&0.579&&0.67\hfil
&&0.79\hfil&&0.536\hfil&&0.390&&0.545 &&\hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 2\hfil&&372.531&&40.58&&0.12$~\pm~$18\%&&0.143&&\hfil
&&\hfil&&0.118\hfil&&0.081&&0.118 &&\hfil&\cr
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3p\ {\rm ^3D} - 3d\ {\rm ^3D^0}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 2\hfil&&300.434&&40.58&&0.22$~\pm~$11\%&&0.254&& \hfil
&& \hfil&&0.288\hfil&&0.285&&0.247 && \hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&3 $-$ 3\hfil&&301.761&&40.58&&0.30$~\pm~$8\%\hfil&&0.320&& \hfil
&&\hfil&&0.362\hfil&&0.355&&0.311 && \hfil&\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&$\rm 3p\ {\rm ^3P} - 3d\ {\rm ^3P^0}$\hfil&& && && && && && && && && && &\cr
\noalign{\hrule}
height4pt&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&&\omit&\cr
&2 $-$ 2\hfil&&344.405&&40.85&&0.28$~\pm~$9\%\hfil&&0.251&& \hfil
&& \hfil&&0.262\hfil&&0.194&& 0.263&& \hfil&\cr
}
\hrule}}
\end{table}